\documentstyle{article}
\begin{document}
\begin{table}%t1
%\centering
\par
  \caption{WHISP early-type disk galaxy sample: basic data. (1)~UGC
  number, (2)~alternative name, (3)~RA, (4)~Dec, (5)~morphological
  type, (6)~absolute $B$-band magnitude, (7)~heliocentric radial
  velocity, (8)~distance, (9)~inclination angle (superscripts
  indicate source: $^t$~tilted ring fits to velocity fields, $^L$~LEDA  
  and $^e$~estimated from our optical data) and (10)~group/cluster
  membership (superscripts indicate source: $^1$~\citet{Garcia93},
  $^2$~\citet{Tully88}, $^3$~\citet{Tully80}, $^4$~\citet{Hickson89}
  and $^5$ this study).  
  Columns (3)--(5) were taken from NED, Col. (6) from LEDA and 
  Cols. (7) and (8) from this study (cf.\ Table~\ref{table:data}).  
\label{table:basicdata}}
  \begin{tabular}{cccccccclccccl}
  \hline \hline 
  \multicolumn{1}{c}{UGC} &
  \multicolumn{1}{c@{\hspace{0.6cm}}}{Alternative} &  
  \multicolumn{3}{c@{\hspace{0.6cm}}}{RA (2000)} &
  \multicolumn{3}{c@{\hspace{0.7cm}}}{Dec (2000)} &  
  \multicolumn{1}{c@{\hspace{0.7cm}}}{Type} &
  \multicolumn{1}{c}{$M_{\rm  B}$} &
  \multicolumn{1}{c}{$V_{\rm {hel}}$} & \multicolumn{1}{c}{$D$} & 
  \multicolumn{1}{c}{$i$} & \multicolumn{1}{c}{\hspace{-1.cm}Group/cluster} \\ 
\par
  & \multicolumn{1}{c@{\hspace{0.6cm}}}{name} & 
  \multicolumn{1}{c@{\hspace{0.3cm}}}{h} &
  \multicolumn{1}{c@{\hspace{0.25cm}}}{m} & 
  %\multicolumn{1}{c@{\hspace{0.7cm}}}{s} & 
  \multicolumn{1}{c@{\hspace{0.2cm}}}{s} & 
  \multicolumn{1}{c@{\hspace{0.35cm}}}{$^{\circ}$} &
  \multicolumn{1}{c@{\hspace{0.2cm}}}{\arcmin} & 
  %\multicolumn{1}{c@{\hspace{0.7cm}}}{\arcsec} & &
  \multicolumn{1}{c@{\hspace{0.3cm}}}{\arcsec} & &
  \multicolumn{1}{c}{mag} & \multicolumn{1}{c}{km~s$^{-1}$} &
  \multicolumn{1}{c}{Mpc} & \multicolumn{1}{c}{\deg} &
  \multicolumn{1}{c}{\hspace{-1.cm}membership} \\       
\par
  \multicolumn{1}{c}{(1)} & \multicolumn{1}{c@{\hspace{0.6cm}}}{(2)} &  
  \multicolumn{3}{c@{\hspace{0.6cm}}}{(3)} &
  \multicolumn{3}{c@{\hspace{0.7cm}}}{(4)} &  
  \multicolumn{1}{c@{\hspace{0.7cm}}}{(5)} & \multicolumn{1}{c}{(6)} & 
  \multicolumn{1}{c}{(7)} & \multicolumn{1}{c}{(8)} &
  \multicolumn{1}{c}{(9)} & \multicolumn{1}{c}{\hspace{-1.cm}(10)}\\ 
  \hline 
\par
  %\endfoot
\par
  89     & NGC 23   & 0  &  9 & 53.4 & 25 & 55 & \hspace{-2.5mm}26 & SB(s)a          & --21.48 &  4555  &  62.1 & 50$^{\rm t}$ & LGG 2$^1$ \\ 
  94     & NGC 26   & 0  & 10 & 25.9 & 25 & 49 & \hspace{-2.5mm}55 & SA(rs)ab 	      & --20.21 &  4589  &  62.6 & 42$^{\rm t}$ & LGG 2$^1$ \\  
  232    & --       & 0  & 24 & 38.7 & 33 & 15 & \hspace{-2.5mm}22 & SB(r)a 	      & --19.96 &  4839  &  66.3 & 48$^{\rm L}$ & -- \\  
  499    & NGC 262  & 0  & 48 & 47.1 & 31 & 57 & \hspace{-2.5mm}25 & SA(s)0/a:       & --20.09 &  4534  &  62.0 & 46$^{\rm L}$ & LGG 14$^1$ \\  
  508    & NGC 266  & 0  & 49 & 47.8 & 32 & 16 & \hspace{-2.5mm}40 & SB(rs)ab 	      & --21.97 &  4647  &  63.6 & 25$^{\rm t}$ & LGG 14$^1$ \\  
  624    & NGC 338  & 1  &  0 & 36.4 & 30 & 40 & \hspace{-2.5mm}8  & Sab 	      & --21.42 &  4772  &  65.1 & 59$^{\rm t}$ & LGG 14$^1$ \\  
  798    & IC 1654  & 1  & 15 & 11.9 & 30 & 11 & \hspace{-2.5mm}41 & (R)SB(r)a       & --19.67 &  4896  &  66.7 & 40$^{\rm t}$ & LGG 18$^1$ \\  
  1310   & NGC 694  & 1  & 50 & 58.5 & 21 & 59 & \hspace{-2.5mm}51 & S0? pec 	      & --19.23 &  2958  &  40.2 & 47$^{\rm e}$ & LGG 34$^1$ \\  
  1541   & NGC 797  & 2  & 3  & 27.9 & 38 & 7  & \hspace{-2.5mm}1  & SAB(s)a 	      & --21.10 &  5649  &  77.0 & 41$^{\rm t}$ & pair with NGC~801$^5$ \\  
  2045   & NGC 972  & 2  & 34 & 13.4 & 29 & 18 & \hspace{-2.5mm}41 & Sab             & --20.39 &  1527  &  21.4 & 61$^{\rm e}$ & 52 --0$^2$ (Cetus) \\ 
  2141   & NGC 1012 & 2  & 39 & 14.9 & 30 & 9  & \hspace{-2.5mm}6  & S0/a?           & --18.02 &  987   &  14.3 & 90$^{\rm e}$ & 17 --3$^2$ (Triang.\ Spur) \\  
  2154   & NGC 1023 & 2  & 40 & 24.0 & 39 & 3  & \hspace{-2.5mm}48 & SB(rs)0-        & --20.23 &  695   &  10.9 & 64$^{\rm L}$ & N 1023 group$^3$ \\  
  2183   & NGC 1056 & 2  & 42 & 48.3 & 28 & 34 & \hspace{-2.5mm}27 & Sa: 	      & --19.60 &  1540  &  21.5 & 90$^{\rm e}$ & --  \\  
  2487   & NGC 1167 & 3  & 1  & 42.4 & 35 & 12 & \hspace{-2.5mm}21 & SA0- 	      & --21.73 &  4950  &  67.4 & 36$^{\rm t}$ & LGG 80$^1$ \\  
  2916   & --       & 4  & 2  & 33.8 & 71 & 42 & \hspace{-2.5mm}21 & Sab 	      & --20.34 &  4518  &  63.5 & 46$^{\rm t}$ & LGG 115$^1$ \\  
  2941   & IC 357   & 4  &  3 & 44.0 & 22 & 9  & \hspace{-2.5mm}33 & SB(s)ab 	      & --21.38 &  6261  &  83.9 & 45$^{\rm e}$ & group with UGC~2942/43$^5$ \\  
  2953   & IC 356   & 4  & 7  & 46.9 & 69 & 48 & \hspace{-2.5mm}45 & SA(s)ab pec     & --21.49 &  894   &  15.1 & 50$^{\rm t}$ & 12 +15$^2$ (U.\ Major) \\  
  3205   & --       & 4  & 56 & 14.8 & 30 &  3 & \hspace{-2.5mm}8  & Sab	      & --21.47 &  3587  &  48.7 & 67$^{\rm t}$ & -- \\  
  3354   & --       & 5  & 47 & 18.2 & 56 &  6 & \hspace{-2.5mm}44 & Sab: 	      & --20.00 &  3084  &  43.6 & 90$^{\rm e}$ & -- \\  
  3382   & --       & 5  & 59 & 47.7 & 62 &  9 & \hspace{-2.5mm}29 & SB(rs)a 	      & --20.62 &  4501  &  62.8 & 16$^{\rm t}$ & -- \\  
  3407   & --       & 6  & 9  & 8.1  & 42 &  5 & \hspace{-2.5mm}7  & Sa 	      & --20.05 &  3606  &  49.8 & 45$^{\rm e}$ & -- \\  
  3426   & --       & 6  & 15 & 36.3 & 71 & 2  & \hspace{-2.5mm}15 & S0: 	      & --20.89 &  4005  &  56.6 & 37$^{\rm e}$ & LGG 135$^1$ \\  
  3546   & NGC 2273 & 6  & 50 & 8.7  & 60 & 50 & \hspace{-2.5mm}45 & SB(r)a 	      & --20.16 &  1838  &  27.3 & 52$^{\rm t}$ & 24 --1$^2$ (Lynx) \\ 
  3580   & --       & 6  & 55 & 30.8 & 69 & 33 & \hspace{-2.5mm}47 & SA(s)a pec:     & --18.19 &  1200  &  19.2 & 63$^{\rm t}$ & 12 --0$^2$ (U.\ Major) \\  
  3642   & --       & 7  & 4  & 20.3 & 64 & 1  & \hspace{-2.5mm}13 & SA0 	      & --21.00 &  4498  &  62.9 & 45$^{\rm t}$ & LGG 140$^1$ \\  
  3965   & IC 2204  & 7  & 41 & 18.1 & 34 & 13 & \hspace{-2.5mm}56 & (R)SB(r)ab      & --18.64 &  4588  &  62.5 & 10$^{\rm e}$ & -- \\  
  3993   & --       & 7  & 55 & 44.0 & 84 & 55 & \hspace{-2.5mm}35 & S0? 	      & --19.61 &  4366  &  61.9 & 20$^{\rm t}$ & pair with UGC~3992$^5$ \\  
  4458   & NGC 2599 & 8  & 32 & 11.3 & 22 & 33 & \hspace{-2.5mm}38 & SAa 	      & --21.12 &  4757  &  64.2 & 27$^{\rm t}$ & pair with PGC~23972$^5$ \\  
  4605   & NGC 2654 & 8  & 49 & 11.9 & 60 & 13 & \hspace{-2.5mm}16 & SBab: sp 	      & --20.08 &  1350  &  20.9 & 90$^{\rm e}$ & 13 --6$^2$ (U. Major S.\ Spur) \\  
  4637   & NGC 2655 & 8  & 55 & 37.7 & 78 & 13 & \hspace{-2.5mm}23 & SAB(s)0/a       & --20.97 &  1415  &  22.4 & 30$^{\rm e}$ & 12 --10$^2$ (U. Major) \\  
  4666   & NGC 2685 & 8  & 55 & 34.7 & 58 & 44 & \hspace{-2.5mm}4  & (R)SB0+ pec     & --19.21 &  876   &  14.6 & 61$^{\rm L}$ & 13 --4$^2$ (U. Major S.\ Spur) \\  
  4862   & NGC 2782 & 9  & 14 & 5.1  & 40 & 6  & \hspace{-2.5mm}49 & SAB(rs)a 	      & --20.99 &  2540  &  35.9 & 20$^{\rm e}$ & 21 --0$^2$ (Leo) \\  
  5060   & NGC 2893 & 9  & 30 & 17.0 & 29 & 32 & \hspace{-2.5mm}24 & (R)SB0/a 	      & --18.14 &  1699  &  24.1 & 24$^{\rm e}$ & 21 --0$^2$ (Leo) \\  
  5253   & NGC 2985 & 9  & 50 & 22.2 & 72 & 16 & \hspace{-2.5mm}43 & (R')SA(rs)ab    & --20.72 &  1325  &  21.1 & 38$^{\rm t}$ & 12 --7$^2$ (U.\ Major) \\  
  5351   & NGC 3067 & 9  & 58 & 21.0 & 32 & 22 & \hspace{-2.5mm}12 & SAB(s)ab?       & --19.35 &  1486  &  21.5 & 70$^{\rm L}$ & 21 --12$^2$ (Leo) \\  
  5559   & NGC 3190 & 10 & 18 & 5.6  & 21 & 49 & \hspace{-2.5mm}55 & SA(s)a pec sp   & --19.88 &  1308  &  18.6 & 67$^{\rm L}$ & HCG 44$^4$ \\  
  5906   & NGC 3380 & 10 & 48 & 12.2 & 28 & 36 & \hspace{-2.5mm}07 & (R')SBa?	      & --18.69 &  1601  &  23.1 & 29$^{\rm e}$ & LGG 227$^1$ \\  
  5960   & NGC 3413 & 10 & 51 & 20.7 & 32 & 45 & \hspace{-2.5mm}59 & S0 	      & --17.39 &   643  &  10.5 & 77$^{\rm t}$ & -- \\  
  6001   & NGC 3442 & 10 & 53 & 8.1  & 33 & 54 & \hspace{-2.5mm}37 & Sa? 	      & --17.77 &  1730  &  25.1 & 47$^{\rm L}$ & 21 --9$^2$ (Leo) \\  
  6118   & NGC 3504 & 11 & 3  & 11.2 & 27 & 58 & \hspace{-2.5mm}21 & (R)SAB(s)ab     & --20.28 &  1536  &  22.2 & 24$^{\rm e}$ & 21 --7$^2$ (Leo) \\  
  6283   & NGC 3600 & 11 & 15 & 52.0 & 41 & 35 & \hspace{-2.5mm}29 & Sa? 	      & --17.17 &   713  &  12.0 & 90$^{\rm e}$ & 15 --9$^2$ (Leo Spur) \\  
  6621   & NGC 3786 & 11 & 39 & 42.5 & 31 & 54 & \hspace{-2.5mm}33 & (R')SAB(r)a pec & --19.74 &  2742  &  38.7 & 58$^{\rm L}$ & 13 --0$^2$ (U. Major S.\ Spur) \\  
  6623   & NGC 3788 & 11 & 39 & 44.6 & 31 & 55 & \hspace{-2.5mm}52 & SAB(rs)ab pec   & --20.24 &  2670  &  37.7 & 74$^{\rm L}$ & 13 --0$^2$ (U. Major S.\ Spur) \\  
\hline
\end{tabular}
\end{table}

\clearpage
  \setcounter{table}{0}
  \begin{table}
  \caption{continued}
  \begin{tabular}{cccccccclccccl}
  \hline \hline 
  \multicolumn{1}{c}{UGC} &
  \multicolumn{1}{c@{\hspace{0.6cm}}}{Alternative} &  
  \multicolumn{3}{c@{\hspace{0.6cm}}}{RA (2000)} &
  \multicolumn{3}{c@{\hspace{0.7cm}}}{Dec (2000)} &  
  \multicolumn{1}{c@{\hspace{0.7cm}}}{Type} &
  \multicolumn{1}{c}{$M_{\rm  B}$} &
  \multicolumn{1}{c}{$V_{\rm {hel}}$} & \multicolumn{1}{c}{$D$} & 
  \multicolumn{1}{c}{$i$} & \multicolumn{1}{c}{\hspace{-1.cm}Group/cluster} \\ 
\par
  & \multicolumn{1}{c@{\hspace{0.6cm}}}{name} & 
  \multicolumn{1}{c@{\hspace{0.3cm}}}{h} &
  \multicolumn{1}{c@{\hspace{0.25cm}}}{m} & 
  %\multicolumn{1}{c@{\hspace{0.7cm}}}{s} & 
  \multicolumn{1}{c@{\hspace{0.2cm}}}{s} & 
  \multicolumn{1}{c@{\hspace{0.35cm}}}{$^{\circ}$} &
  \multicolumn{1}{c@{\hspace{0.2cm}}}{\arcmin} & 
  %\multicolumn{1}{c@{\hspace{0.7cm}}}{\arcsec} & &
  \multicolumn{1}{c@{\hspace{0.3cm}}}{\arcsec} & &
  \multicolumn{1}{c}{mag} & \multicolumn{1}{c}{km~s$^{-1}$} &
  \multicolumn{1}{c}{Mpc} & \multicolumn{1}{c}{\deg} &
  \multicolumn{1}{c}{\hspace{-1.cm}membership} \\       
\par
  \multicolumn{1}{c}{(1)} & \multicolumn{1}{c@{\hspace{0.6cm}}}{(2)} &  
  \multicolumn{3}{c@{\hspace{0.6cm}}}{(3)} &
  \multicolumn{3}{c@{\hspace{0.7cm}}}{(4)} &  
  \multicolumn{1}{c@{\hspace{0.7cm}}}{(5)} & \multicolumn{1}{c}{(6)} & 
  \multicolumn{1}{c}{(7)} & \multicolumn{1}{c}{(8)} &
  \multicolumn{1}{c}{(9)} & \multicolumn{1}{c}{\hspace{-1.cm}(10)}\\ 
  \hline 
\par
  %\endfoot
\par
  6742   & NGC 3870 & 11 & 45 & 56.6 & 50 & 12 & \hspace{-2.5mm}00 & S0?             & --17.28 &   752  &  13.0 & 36$^{\rm L}$ & 12 --1$^2$ (U.\ Major) \\  
  6786   & NGC 3900 & 11 & 49 & 9.4  & 27 & 1  & \hspace{-2.5mm}19 & SA(r)0+ 	      & --19.94 &  1799  &  25.9 & 65$^{\rm t}$ & 13 --9$^2$ (U. Major S.\ Spur) \\  
  6787   & NGC 3898 & 11 & 49 & 15.4 & 56 & 5  & \hspace{-2.5mm}4  & SA(s)ab 	      & --20.25 &  1171  &  18.9 & 67$^{\rm t}$ & 12 --3$^2$ (U.\ Major) \\  
  7166   & NGC 4151 & 12 & 10 & 32.6 & 39 & 24 & \hspace{-2.5mm}21 & (R')SAB(rs)ab:  & --20.84 &   998  &  15.9 & 20$^{\rm t}$ & 12 --6$^2$ (U.\ Major) \\  
  7256   & NGC 4203 & 12 & 15 & 5.0  & 33 & 11 & \hspace{-2.5mm}50 & SAB0-: 	      & --19.47 &  1091  &  16.9 & 40$^{\rm t}$ & 14 --1$^2$ (Coma--Sculptor) \\  
  7489   & NGC 4369 & 12 & 24 & 36.2 & 39 & 22 & \hspace{-2.5mm}59 & (R)SA(rs)a      & --18.87 &  1027  &  16.4 & 17$^{\rm L}$ & 12 --6$^2$ (U.\ Major) \\  
  7506   & NGC 4384 & 12 & 25 & 12.0 & 54 & 30 & \hspace{-2.5mm}22 & Sa              & --19.70 &  2532  &  37.0 & 41$^{\rm L}$ & 42 +11$^2$ (Canes Venatici) \\  
  7704   & NGC 4509 & 12 & 33 &  6.8 & 32 & 5  & \hspace{-2.5mm}30 & Sab pec? 	      & --16.17 &  937   &  14.8 & 60$^{\rm e}$ & LGG 279$^1$ \\  
  7989   & NGC 4725 & 12 & 50 & 26.6 & 25 & 30 & \hspace{-2.5mm}3  & SAB(r)ab pec    & --21.69 &  1208  &  18.2 & 51$^{\rm t}$ & 14 --2$^2$ (Coma--Sculptor) \\  
  8271   & NGC 5014 & 13 & 11 & 31.2 & 36 & 16 & \hspace{-2.5mm}55 & Sa? sp 	      & --18.44 &  1128  &  17.7 & 60$^{\rm e}$ & 43 --1$^2$ (Canes V.\ Spur)\\
  8699   & NGC 5289 & 13 & 45 & 8.7  & 41 & 30 & \hspace{-2.5mm}12 & (R)SABab:	      & --19.70 &  2521  &  36.7 & 72$^{\rm t}$ & LGG 361$^1$ \\  
  8805   & NGC 5347 & 13 & 53 & 17.8 & 33 & 29 & \hspace{-2.5mm}27 & (R')SB(rs)ab    & --19.69 &  2384  &  34.5 & 30$^{\rm e}$ & 42 --0$^2$ (Canes Venatici) \\  
  8863   & NGC 5377 & 13 & 56 & 16.7 & 47 & 14 & \hspace{-2.5mm}8  & (R)SB(s)a       & --20.32 &  1791  &  27.2 & 51$^{\rm t}$ & 42 --0$^2$ (Canes Venatici) \\  
  9133   & NGC 5533 & 14 & 16 & 7.7  & 35 & 20 & \hspace{-2.5mm}38 & SA(rs)ab 	      & --21.47 &  3861  &  54.3 & 53$^{\rm t}$ & LGG 380$^1$ \\  
  9644   & --       & 14 & 59 & 34.3 & 27 & 6  & \hspace{-2.5mm}58 & SB(r)a 	      & --19.25 &  6665  &  91.5 & 19$^{\rm e}$ & -- \\  
  10448  & NGC 6186 & 16 & 34 & 25.5 & 21 & 32 & \hspace{-2.5mm}27 & (R')SB(s)a      & --21.65 &  11351 & 153.8 & 30$^{\rm e}$ & -- \\  
  11269  & NGC 6667 & 18 & 30 & 39.8 & 67 & 59 & \hspace{-2.5mm}13 & SABab? pec      & --20.11 &  2581  &  38.3 & 56$^{\rm L}$ & 70 --0$^2$ (--) \\  
  11670  & NGC 7013 & 21 & 3  & 33.6 & 29 & 53 & \hspace{-2.5mm}51 & SA(r)0/a 	      & --19.53 &  775   &  12.7 & 68$^{\rm t}$ & 65 +5$^2$ (Pegasus Spur) \\  
  11852  & --       & 21 & 55 & 59.3 & 27 & 53 & \hspace{-2.5mm}54 & SBa? 	      & --20.25 &  5846  &  80.0 & 50$^{\rm t}$ & -- \\  
  11914  & NGC 7217 & 22 & 7  & 52.4 & 31 & 21 & \hspace{-2.5mm}33 & (R)SA(r)ab      & --20.45 &  949   &  14.9 & 31$^{\rm t}$ & 65 +1$^2$ (Pegasus Spur) \\  
  11951  & NGC 7231 & 22 & 12 & 30.1 & 45 & 19 & \hspace{-2.5mm}42 & SBa 	      & --19.14 &  1086  &  17.4 & 70$^{\rm L}$ & -- \\  
  12043  & NGC 7286 & 22 & 27 & 50.5 & 29 & 5  & \hspace{-2.5mm}45 & S0/a 	      & --17.15 &  1007  &  15.4 & 67$^{\rm t}$ & -- \\  
  12276  & NGC 7440 & 22 & 58 & 32.5 & 35 & 48 & \hspace{-2.5mm}9  & SB(r)a 	      & --20.59 &  5662  &  77.7 & 37$^{\rm L}$ & -- \\  
  12713  & --       & 23 & 38 & 14.4 & 30 & 42 & \hspace{-2.5mm}29 & S0/a 	      & --14.35 &   295  &   5.7 & 72$^{\rm t}$ & 65 +4$^2$ (Pegasus Spur) \\  
  12815  & NGC 7771 & 23 & 51 & 24.9 & 20 & 6  & \hspace{-2.5mm}43 & SB(s)a 	      & --21.60 &  4307  &  58.5 & 61$^{\rm L}$ & LGG 483$^1$ \\\hline 
%\end{longtable}
\end{tabular}
\end{table}

\clearpage
\begin{table}%t2
  \caption{Observational parameters: (1) UGC number, (2) observation
  date, (3) spatial resolution of the unsmoothed data, (4) velocity
  resolution before Hanning smoothing, (5) rms noise level in the
  unsmoothed channel maps, (6) 1 sigma column density in the \HI\
  maps shown in the atlas, (7) previous 21~cm synthesis observations
  (references are explained at the bottom of the table) and (8)
  remarks. \label{table:observations}}   
  \begin{tabular}{ccccccccc}
  \hline \hline
  \multicolumn{1}{c}{UGC} & Observation date &
  \multicolumn{3}{c}{Resolution} & $\sigma_{\rm {ch}}$ & 
  $\sigma_{\rm {map}}$ & Previous & Remarks \\    
  & & RA & Dec & $v$ \hspace{0.22cm}  & & & observations & \\ 
    & & \multicolumn{1}{c}{\arcsec} & \multicolumn{1}{l}{\arcsec} & 
  km~s$^{-1}$\hspace{-0.2cm} & mJy/beam & $10^{19}$ atoms~cm$^{-2}$ &
  & \\  
    (1) & (2) & \multicolumn{2}{c}{(3)} & (4)\hspace{0.2cm} & (5) & (6)
  & \hspace{0.35cm}(7) & \hspace{0.3cm} (8) \\   
  \hline
  %\endfirsthead  
  %% This comes at the head of the table on the following pages
  %\hline
  %\endhead  
 %
  %% This comes at the end of each page
   %
  %\endfoot
 %
  %% This comes at the end of the last page
    89    & Sep. 2000 	      & 16.1  & 37.9  & 10.03 & 0.64 & 3.25 & O93      & \rdelim\}{2}{2cm}[1 pointing] \\
  94    & Sep. 2000	      & 16.1  & 37.9  & 10.03 & 0.64 & 2.86 & O93      & \\      
  232   & July 2000 	      & 15.7  & 29.9  & 10.04 & 0.78 & 4.97 & --       & \\
  499   & Aug. 2000 	      & 11.1  & 19.4  & 10.04 & 0.69 & 2.56 & H82, S87 & \rdelim\}{2}{2cm}[1 pointing] \\
  508   & Aug. 2000 	      & 11.1  & 19.4  & 10.04 & 0.69 & 2.82 & --       & \\
  624   & July 2000 	      & 10.7  & 24.2  & 10.04 & 0.61 & 11.3 & --       & \\
  798   & Aug. 2000 	      & 12.6  & 23.2  & 10.04 & 0.73 & 3.13 & --       & \\
  1310 	& June 2000 	      & 12.0  & 31.8  & 9.98  & 0.90 & 3.66 & VM88     & \\
  1541 	& Aug. 2000 	      & 16.1  & 27.0  & 10.07 & 0.79 & 7.06 & --       & \\
  2045 	& Aug. 2000  	      & 15.9  & 34.0  & 9.94  & 0.67 & 2.75 & H03       & \\
  2141 	& Jan. 1994$^\dagger$ & 8.3   & 19.4  & 4.98  & 3.51 & 71.8 & --       & \\
  2154 	& July 1997$^\dagger$ & 10.1  & 18.0  & 19.80 & 1.20 & 2.88 & A79, S84 & \\
  2183 	& Apr. 2000 	      & 10.6  & 26.0  & 9.94  & 0.79 & 6.52 & --       & \\
  2487 	& Dec. 2000 	      & 11.8  & 22.2  & 10.06 & 0.87 & 1.29 & --       & \\
  2916 	& Apr. 2000 	      & 10.8  & 13.2  & 10.03 & 0.59 & 16.8 & --       & \\
  2941 	& July 2000 	      & 10.7  & 32.3  & 5.06  & 1.31 & 6.47 & --       & \\
  2953 	& Sep. 2001 	      & 10.7  & 14.4  & 9.91  & 0.60 & 1.39 & --       & \\
  3205  & Dec. 2000 	      & 10.6  & 24.5  & 10.01 & 0.61 & 15.5 & --       & \\
  3354  & Nov. 2000 	      & 16.9  & 21.1  & 9.98  & 0.57 & 5.93 & --       & \\
  3382  & Dec. 2000 	      & 12.2  & 14.1  & 5.04  & 0.75 & 1.20 & --       & \\
  3407  & Dec. 2000 	      & 12.7  & 17.8  & 10.01 & 0.71 & 2.38 & --       & \\
  3426  & Jan. 2002 	      & 12.3  & 13.0  & 5.03  & 0.54 & 1.20 & --       & \\
  3546  & Dec. 2000 	      & 12.2  & 14.9  & 9.95  & 0.63 & 1.13 & VW83, VD91     & \\
  3580  & Jan. 1994$^\dagger$ & 9.1   & 10.2  & 4.98  & 2.50 & 5.45 & B94      & \\
  3642  & May 2000  	      & 12.6  & 13.4  & 10.03 & 0.75 & 3.69 & --       & \\
  3965  & Apr. 2000 	      & 11.7  & 21.4  & 2.51  & 1.84 & 1.95 & --       & \\
  3993  & July 2000 	      & 12.2  & 12.3  & 5.03  & 0.82 & 1.32 & --       & \\
  4458  & July 2002 	      & 15.5  & 43.6  & 10.04 & 0.74 & 3.14 & --       & \\
  4605  & Apr. 2000 	      & 12.8  & 13.7  & 9.94  & 0.69 & 18.4 & W88      & \\
  4637  & July 2002 	      & 12.6  & 12.4  & 9.94  & 0.68 & 0.74 & --       & \\
  4666  & Aug. 2000 	      & 11.8  & 15.6  & 9.91  & 0.69 & 2.95 & S80      & \\
  4862  & July 2000 	      & 15.5  & 24.8  & 5.00  & 0.99 & 6.07 & S94      & \\
  5060  & July 2000 	      & 15.6  & 33.2  & 4.99  & 0.88 & 2.04 & --       & \\
  5253  & Aug. 1993$^\dagger$ & 11.6  & 12.4  & 19.92 & 1.82 & 3.64 & O93      & \\
  5351  & Aug. 2000 	      & 14.0  & 20.8  & 9.94  & 0.75 & 4.11 & C89, C92 & \\
  5559  & June 1996$^\dagger$ & 9.8   & 26.3  & 19.92 & 1.98 & 31.1 & W91      & \\
  5906  & Aug. 2000 	      & 34.6  & 15.8  & 4.99  & 1.05 & 2.04 & --       & \\
  5960  & Apr. 2000 	      & 21.7  & 11.6  & 4.97  & 1.40 & 6.45 & --       & \\
  6001  & Aug. 2000 	      & 21.4  & 11.7  & 4.99  & 0.97 & 5.07 & --       & \\
  6118  & July 2000 	      & 35.2  & 15.7  & 9.94  & 0.78 & 4.35 & VM83b    & \\
  6283  & Apr. 1996$^\dagger$ & 18.7  & 12.5  & 4.97  & 3.11 & 28.9 & GR02     & \\
\hline
\end{tabular}
\end{table}

\clearpage
  \setcounter{table}{1}
  \begin{table}
  \caption{continued}
  \begin{tabular}{ccccccccc}
  \hline \hline
  \multicolumn{1}{c}{UGC} & Observation date &
  \multicolumn{3}{c}{Resolution} & $\sigma_{\rm {ch}}$ & 
  $\sigma_{\rm {map}}$ & Previous & Remarks \\    
  & & RA & Dec & $v$ \hspace{0.22cm}  & & & observations & \\ 
    & & \multicolumn{1}{c}{\arcsec} & \multicolumn{1}{l}{\arcsec} & 
  km~s$^{-1}$\hspace{-0.2cm} & mJy/beam & $10^{19}$ atoms~cm$^{-2}$ &
  & \\  
    (1) & (2) & \multicolumn{2}{c}{(3)} & (4)\hspace{0.2cm} & (5) & (6)
  & \hspace{0.35cm}(7) & \hspace{0.3cm} (8) \\   
  \hline
  6621  & July 2002 	      & 31.2  & 15.9  & 9.97  & 0.59 & 3.70 & O93      & \rdelim\}{2}{2cm}[1 pointing] \\
  6623  & July 2002 	      & 31.2  & 15.9  & 9.97  & 0.59 & 3.70 & O93      & \\
  6742  & Apr. 2000 	      & 14.1  & 12.8  & 4.97  & 0.81 & 1.84 & --       & \\
  6786  & Aug. 2000 	      & 15.5  & 36.0  & 9.95  & 0.74 & 1.48 & VD89, H00& \\
  6787  & Apr. 2000 	      & 12.6  & 13.8  & 9.92  & 0.65 & 2.14 & VD94     & \\
  7166  & June 1997$^\dagger$ & 20.8  & 12.9  & 4.98  & 2.98 & 1.73 & B77, P92, M99& \\ 
  7256  & July 2000 	      & 30.7  & 15.3  & 4.98  & 1.76 & 2.29 & VD88     & \\
  7489  & Aug. 1997$^\dagger$ & 18.2  & 11.4  & 4.98  & 4.30 & 5.18 & --       & \\
  7506  & Aug. 2000 	      & 13.0  & 10.3  & 5.00  & 0.93 & 13.1 & --       & \\
  7704  & Aug. 2000 	      & 21.3  & 10.0  & 4.98  & 0.93 & 4.69 & --       & \\
  7989  & Oct. 1994$^\dagger$ & 8.0   & 13.2  & 19.80 & 1.13 & 9.73 & W84      & \\
  8271  & July 2000 	      & 10.2  & 19.9  & 4.98  & 0.91 & 0.98 & --       & \\
  8699  & Aug. 2000 	      & 16.9  & 10.7  & 9.97  & 0.68 & 12.4 & VM83a    & \\
  8805  & July 1997$^\dagger$ & 19.5  & 10.6  & 5.00  & 4.41 & 2.49 & --       & \\
  8863  & Aug. 2000 	      & 15.8  & 22.4  & 9.95  & 0.66 & 2.11 & --       & \\
  9133  & July 2000 	      & 15.8  & 28.8  & 10.02 & 0.70 & 2.44 & B92, B94 & \\
  9644  & July 2000 	      & 10.5  & 25.6  & 5.08  & 1.08 & 9.41 & --       & \\
  10448 & July 2000 	      & 10.6  & 32.1  & 5.15  & 1.10 & 5.26 & --       & \\
  11269 & July 2000 	      & 12.2  & 14.4  & 9.97  & 0.59 & 1.78 & --       & \\
  11670 & May 2000 	      & 11.3  & 25.0  & 9.91  & 0.73 & 2.37 & K84      & \\
  11852 & July 2000 	      & 16.1  & 35.7  & 10.08 & 0.78 & 4.12 & --       & \\
  11914 & Sep. 2000 	      & 10.5  & 23.5  & 9.92  & 0.68 & 2.25 & VM95     & \\
  11951 & Jan. 1994$^\dagger$ & 10.1  & 14.1  & 4.98  & 3.04 & 54.5 & --       & \\
  12043 & Apr. 2000 	      & 9.9   & 25.9  & 4.98  & 1.22 & 9.50 & --       & \\
  12276 & July 2002 	      & 10.4  & 19.3  & 5.05  & 0.84 & 9.45 & --       & \\
  12713 & June 2000 	      & 11.6  & 22.0  & 4.97  & 0.95 & 1.78 & VD91b     & \\
  12815 & Aug. 2000           & 10.4  & 33.4  & 10.03 & 0.70 & 9.62 & --       & \\
\hline          	      
\end{tabular}
 \medskip
 {$^\dagger$ Observed before the upgrade of
  the WSRT.} \\%[0.25cm]
       {Explanation of the references in col.~(7):} \\%[0.1cm]
A79  {\citet{Allsopp79}}; 
B77  {\citet{Bosma77}}; 
B92  {\citet{Broeils92}};
B94  {\citet{Broeils94}};
C89  {\citet{Carilli89}};
C92  {\citet{Carilli92}};  
GR02 {\citet{Garcia-Ruiz02}};
H00  {\citet{Haynes00}};
H03 {\citet{Hameed03}};
H82  {\citet{Heckman82}};
K84  {\citet{Knapp84}};
M99  {\citet{Mundell99}};
O93  {\citet{Oosterloo93}};
P92  {\citet{Pedlar92}};  
S80  {\citet{Shane80}};
S84  {\citet{Sancisi84}};   
{S87}  {\citet{Simkin87}};           
{S94} {\citet{Smith94}};             
{VD88} {\citet{VanDriel88}};     
{VD89} {\citet{VanDriel89}};      
{VD91} {\citet{VanDriel91b}};
{VD91b} {\citet{VanDriel91}};      
{VD94} {\citet{VanDriel94}};     
{VM83a} {\citet{VanMoorsel83}};        
{VM83b} {\citet{VanMoorsel83b}};        
{VM88} {\citet{VanMoorsel88}};         
{VM95} {\citet{Verdes-Montenegro95}}; 
{VW83} {\citet{VanWoerden83}};
{W84}  {\citet{Wevers84}};   
{W88}  {\citet{Warmels88b}};                     
{W91}  {\citet{Williams91}}.  
  %\endlastfoot	     	
\end{table}


\clearpage
\begin{table}%t3
  \caption{\HI\ properties: (1) UGC number, (2) systemic velocity, (3) 
  distance, (4) line width of global profile at the 20\% level, (5)
  idem at the 50\% level, (6) total \HI\ flux, (7) total \HI\ mass,
  (8) \HI\ radius in \arcsec, (9) idem in kpc, (10) average face-on
  \HI\ surface density within \HI\ radius, (11) idem within optical
  radius, (12) amount of asymmetry in global profile, (13)
  morphological asymmetry, (14) kinematical asymmetry, and (15)
  evidence for current interaction/merging/accretion.  
  \label{table:data}}
  %\begin{center}
\begin{tabular}{ccccccccccccccc}
  \noalign{\smallskip}
  \hline \hline\noalign{\smallskip}
  \multicolumn{1}{c}{UGC} & \multicolumn{1}{c}{$V_{\rm {sys}}$} & 
  \multicolumn{1}{c}{$D$} & \multicolumn{1}{c}{$W_{20}^c$} &
  \multicolumn{1}{c}{$W_{50}^c$} & \multicolumn{1}{c}{$\int
  F~\rm {d}v$} & \multicolumn{1}{c}{$M_{\rm HI}$} &
  \multicolumn{1}{c}{${R}_{\rm HI}$} & \multicolumn{1}{c}{${R}_{\rm HI}$} & \multicolumn{1}{c}{$\langle\Sigma_{\rm HI}\rangle_{{R}_{\rm HI}}$}
  & \multicolumn{1}{c}{$\langle\Sigma_{\rm HI}\rangle_{{R}_{25}}$} &
  \multicolumn{4}{c}{Asymmetries} \\   
  & \multicolumn{1}{c}{km~s$^{-1}$} & \multicolumn{1}{c}{Mpc} &
  \multicolumn{1}{c}{km~s$^{-1}$} & \multicolumn{1}{c}{km~s$^{-1}$} &
  \multicolumn{1}{c}{Jy~km~s$^{-1}$} & \multicolumn{1}{c}{10$^9$ ${M}_{\odot}$} &
   \multicolumn{1}{c}{\arcsec} & \multicolumn{1}{c}{kpc} &
  \multicolumn{1}{c}{${M}_{\odot}$} & \multicolumn{1}{c}{${M}_{\odot}$} &
  prof. & morph. & kin. & interac. \\   
  \multicolumn{1}{c}{(1)} & \multicolumn{1}{c}{(2)} &
  \multicolumn{1}{c}{(3)} & \multicolumn{1}{c}{(4)} &
  \multicolumn{1}{c}{(5)} & \multicolumn{1}{c}{(6)} &
  \multicolumn{1}{c}{(7)} & \multicolumn{1}{c}{(8)} & 
  \multicolumn{1}{c}{(9)} & \multicolumn{1}{c}{(10)} & 
  \multicolumn{1}{c}{(11)} & \multicolumn{1}{c}{(12)} & 
  \multicolumn{1}{c}{(13)} & \multicolumn{1}{c}{(14)} &
  \multicolumn{1}{c}{(15)} \\  
  \hline  
 89     & 4555  &  62.1 & 440  & 386  & 9.44   	  & 8.63    	   & 77		& 24	  & 3.5    \hspace{0cm} & 5.1\hspace{0cm} & *  	& *	& 	& +   	\\
  94     & 4589  &  62.6 & 319  & 300  & 12.19  	  & 11.28    	   & 85 	& 26 	  & 4.0    \hspace{0cm} & 7.2\hspace{0cm} & *  	& *	& 	& +   	\\
  232    & 4839  &  66.3 & 276  & 236  & 7.66   	  & 7.95    	   & 82 	& 27 	  & 2.1    \hspace{0cm} & 3.3\hspace{0cm} &   	& *	& 	&    	\\
  499    & 4534  &  62.0 &  91  &  65  & 23.9   	  & 21.45    	   & 104 	& 31 	  & 1.6    \hspace{0cm} & 1.8\hspace{0cm} &   	& **	& n.a.	& +   	\\
  508    & 4647  &  63.6 & 478  & 453  & 7.71   	  & 7.40    	   & 94 	& 29 	  & 2.0    \hspace{0cm} & 2.0\hspace{0cm} &   	& *	& 	& +   	\\
  624    & 4772  &  65.1 & 560  & 518  & 14.39  	  & 14.42   	   & 92 	& 29 	  & 4.3    \hspace{0cm} & 7.7\hspace{0cm} &   	& **	& ***	& +   	\\
  798    & 4896  &  66.7 & 220  & 205  & 3.88   	  & 4.08    	   & 72 	& 23 	  & 1.8    \hspace{0cm} & 2.4\hspace{0cm} &   	& 	& 	&    	\\
  1310   & 2958  &  40.2 & 186  & 117  & 6.16   	  & 2.33    	   & 50 	& 9.7	  & 4.8    \hspace{0cm} &14.3\hspace{0cm} & **      & n.a.	& n.a.	& +   	\\
  1541   & 5649  &  77.0 & 446  & 421  & 7.94   	  & 11.14    	   & 75 	& 28 	  & 2.8    \hspace{0cm} & 3.9\hspace{0cm} & ***     & *	& *	&    	\\
  2045   & 1527  &  21.4 & 334  & 289  & 18.89  	  & 2.09    	   & 94 	& 9.9 	  & 4.2    \hspace{0cm} & 3.6\hspace{0cm} &    	&  	& 	&    	\\
  2141   & 987   &  14.3 & 232  & 201  & 47.94  	  & 2.30    	   & 155 	& 10	  & 4.5    \hspace{0cm} &11.3\hspace{0cm} &    	& *	& 	&    	\\
  2154   & 695   &  10.9 & 529  & 497  & 80.13  	  & 2.26    	   & --$^\#$	&--$^\#$  & --$^\#$\hspace{0cm} & 0.7\hspace{0cm} & n.a.    & n.a.	& n.a. 	& +  	\\
  2183   & 1540  &  21.5 & 292  & 268  & 32.99  	  & 3.63   	   & 164 	& 17 	  & 3.0    \hspace{0cm} & 6.8\hspace{0cm} & *  	& *	& *	&    	\\
  2487   & 4950  &  67.4 & 466  & 437  & 15.93  	  & 17.09    	   & 120 	& 39 	  & 1.6    \hspace{0cm} & 1.8\hspace{0cm} &   	& 	& 	&    	\\
  2916   & 4518  &  63.5 & 357  & 337  & 23.08  	  & 21.94    	   & 108 	& 34 	  & 4.3    \hspace{0cm} & 7.3\hspace{0cm} & *  	& 	& **	& +   	\\
  2941   & 6261  &  83.9 & 160  & 139  & 9.84   	  & 16.32    	   & 75 	& 30 	  & 3.4    \hspace{0cm} & 5.4\hspace{0cm} &   	& *	& *	& +   	\\
  2953   & 894   &  15.1 & 484  & 464  & 119.73 	  & 6.43    	   & 300 	& 22 	  & 1.9    \hspace{0cm} & 2.2\hspace{0cm} &    	& 	& 	&    	\\
  3205   & 3587  &  48.7 & 437  & 419  & 16.38  	  & 9.18    	   & 118 	& 28 	  & 3.4    \hspace{0cm} & 4.7\hspace{0cm} &   	& 	& 	&    	\\
  3354   & 3084  &  43.6 & 423  & 398  & 17.71  	  & 7.95    	   & 77 	& 16 	  & 2.0    \hspace{0cm} & 2.3\hspace{0cm} & *       & 	& 	&    	\\
  3382   & 4501  &  62.8 & 204  & 193  & 5.68   	  & 5.27    	   & 78 	& 24 	  & 2.0    \hspace{0cm} & 2.8\hspace{0cm} &   	& 	& 	&    	\\
  3407   & 3606  &  49.8 & 316  & 298  & 3.13   	  & 1.82   	   & 50 	& 12 	  & 2.6    \hspace{0cm} & 3.1\hspace{0cm} & *       & 	& 	& +   	\\
  3426   & 4005  &  56.6 & 161  & 114  & 3.27   	  & 2.53    	   & --$^\#$ 	&--$^\#$  & --$^\#$\hspace{0cm} & 0.2\hspace{0cm} & *       & n.a.	& n.a.	& +   	\\
  3546   & 1838  &  27.3 & 367  & 347  & 13.74  	  & 2.42    	   & 132 	& 18 	  & 2.0    \hspace{0cm} & 2.2\hspace{0cm} &   	& 	& 	&    	\\
  3580   & 1200  &  19.2 & 241  & 224  & 43.24  	  & 3.76    	   & 199 	& 18 	  & 3.0    \hspace{0cm} & 5.4\hspace{0cm} & *  	& 	& * 	&    	\\
  3642   & 4498  &  62.9 & 469  & 441  & 38.13  	  & 35.62    	   & 171 	& 52 	  & 3.1    \hspace{0cm} & 4.7\hspace{0cm} & *       &  	& 	&    	\\
  3965   & 4588  &  62.5 &  93  &  74  & 15.67  	  & 14.47    	   & 106 	& 33 	  & 3.2    \hspace{0cm} & 5.3\hspace{0cm} &   	&  	& 	&    	\\
  3993   & 4366  &  61.9 & 209  & 192  & 7.73   	  & 6.99    	   & 82 	& 24 	  & 1.6    \hspace{0cm} & 1.7\hspace{0cm} &   	& *	& 	&    	\\
  4458   & 4757  &  64.2 & 284  & 245  & 12.24  	  & 11.82   	   & 110 	& 34 	  & 2.0    \hspace{0cm} & 3.1\hspace{0cm} &   	& **	& 	& +   	\\
  4605   & 1350  &  20.9 & 432  & 392  & 58.03  	  & 5.99    	   & 277 	& 28 	  & 1.8    \hspace{0cm} & 3.0\hspace{0cm} &   	& 	& 	&    	\\
  4637   & 1415  &  22.4 & 399  & 243  & 22.34  	  & 2.62    	   & --$^\#$ 	&--$^\#$  & --$^\#$\hspace{0cm} & 0.9\hspace{0cm} & **      & ***	& n.a.	& +   	\\
  4666   & 876   &  14.6 & 303  & 284  & 29.07  	  & 1.47    	   & 184 	& 12 	  & 2.2    \hspace{0cm} & 2.5\hspace{0cm} & *       & *	& 	&    	\\
  4862   & 2540  &  35.9 & 192  & 156  & 16.42  	  & 5.07    	   & 90 	& 16 	  & 3.1    \hspace{0cm} & 2.2\hspace{0cm} & ***     & ***	& n.a.	& +   	\\
  5060   & 1699  &  24.1 & 185  &  86  & 5.13   	  & 0.70   	   & 60		& 7.0	  & 2.3    \hspace{0cm} & 3.7\hspace{0cm} & **      & ***	& 	&    	\\
  5253   & 1325  &  21.1 & 320  & 296  & 132.4  	  & 13.92          & 330 	& 34 	  & 2.1    \hspace{0cm} & 3.8\hspace{0cm} & *       & ***	& 	& +   	\\
  5351   & 1486  &  21.5 & 276  & 187  & 9.23   	  & 0.99    	   & 61 	& 6.3 	  & 3.4    \hspace{0cm} & 3.1\hspace{0cm} & **      & **	& **	& +   	\\
  5559   & 1308  &  18.6 & 473  & 449  & 7.32   	  & 0.60    	   & 88 	& 7.7 	  & 2.6    \hspace{0cm} & 1.5\hspace{0cm} &   	& n.a.	& n.a.	& +   	\\
  5906   & 1601  &  23.1 & 123  & 110  & 2.53   	  & 0.31    	   & 47 	& 5.3 	  & 2.0    \hspace{0cm} & 1.9\hspace{0cm} &   	& *	& 	&    	\\
  5960   &  643  &  10.5 & 180  & 154  & 28.42  	  & 0.75    	   & 109	& 5.7 	  & 6.4    \hspace{0cm} &15.9\hspace{0cm} & **      & *	& **	&    	\\
  6001   & 1730  &  25.1 & 153  & 121  & 3.58   	  & 0.33    	   & 44 	& 5.3	  & 4.2    \hspace{0cm} & 9.6\hspace{0cm} & *       & n.a.	& n.a.	&    	\\
  6118   & 1536  &  22.2 & 219  & 194  & 7.20   	  & 0.84    	   & 89 	& 9.5 	  & 2.7    \hspace{0cm} & 3.0\hspace{0cm} & *  	& 	& 	&    	\\
  6283   &  713  &  12.0 & 216  & 200  & 57.38  	  & 2.00    	   & 204 	& 12 	  & 3.1    \hspace{0cm} & 5.5\hspace{0cm} &   	& *	& 	&    	\\
  6621   & 2742  &  38.7 & 420  & 375  & 3.69$^\dagger$   & 1.30$^\dagger$ & 57 	& 10 	  & 4.1    \hspace{0cm} & 4.1\hspace{0cm} &    	& ***	& n.a.	& +   	\\
\hline
\end{tabular}
\end{table}

\clearpage
\setcounter{table}{2}
\begin{table}
\caption{continued}
\begin{tabular}{ccccccccccccccc}
  \noalign{\smallskip}
  \hline \hline\noalign{\smallskip}
  \multicolumn{1}{c}{UGC} & \multicolumn{1}{c}{$V_{\rm {sys}}$} & 
  \multicolumn{1}{c}{$D$} & \multicolumn{1}{c}{$W_{20}^c$} &
  \multicolumn{1}{c}{$W_{50}^c$} & \multicolumn{1}{c}{$\int
  F~\rm {d}v$} & \multicolumn{1}{c}{$M_{\rm HI}$} &
  \multicolumn{1}{c}{${R}_{\rm HI}$} & \multicolumn{1}{c}{${R}_{\rm HI}$} & \multicolumn{1}{c}{$\langle\Sigma_{\rm HI}\rangle_{{R}_{\rm HI}}$}
  & \multicolumn{1}{c}{$\langle\Sigma_{\rm HI}\rangle_{{R}_{25}}$} &
  \multicolumn{4}{c}{Asymmetries} \\   
  & \multicolumn{1}{c}{km~s$^{-1}$} & \multicolumn{1}{c}{Mpc} &
  \multicolumn{1}{c}{km~s$^{-1}$} & \multicolumn{1}{c}{km~s$^{-1}$} &
  \multicolumn{1}{c}{Jy~km~s$^{-1}$} & \multicolumn{1}{c}{10$^9$ ${M}_{\odot}$} &
   \multicolumn{1}{c}{\arcsec} & \multicolumn{1}{c}{kpc} &
  \multicolumn{1}{c}{${M}_{\odot}$} & \multicolumn{1}{c}{${M}_{\odot}$} &
  prof. & morph. & kin. & interac. \\   
  \multicolumn{1}{c}{(1)} & \multicolumn{1}{c}{(2)} &
  \multicolumn{1}{c}{(3)} & \multicolumn{1}{c}{(4)} &
  \multicolumn{1}{c}{(5)} & \multicolumn{1}{c}{(6)} &
  \multicolumn{1}{c}{(7)} & \multicolumn{1}{c}{(8)} & 
  \multicolumn{1}{c}{(9)} & \multicolumn{1}{c}{(10)} & 
  \multicolumn{1}{c}{(11)} & \multicolumn{1}{c}{(12)} & 
  \multicolumn{1}{c}{(13)} & \multicolumn{1}{c}{(14)} &
  \multicolumn{1}{c}{(15)} \\  
  \hline  
  6623   & 2670  &  37.7 & 520  & 438  & 11.51$^\dagger$  & 3.95$^\dagger$ & 79 	& 15 	  & 3.5    \hspace{0cm} & 4.4\hspace{0cm} & ***     & ***	& n.a.	& +   	\\
  6742   &  752  &  13.0 & 127  &  84  & 4.08   	  & 0.17    	   & 51 	& 3.2 	  & 4.3    \hspace{0cm} & 8.0\hspace{0cm} & *  	& **	& 	&    	\\
  6786   & 1799  &  25.9 & 445  & 423  & 24.91  	  & 3.93    	   & 143 	& 18 	  & 2.4    \hspace{0cm} & 3.4\hspace{0cm} &   	& 	& 	&    	\\
  6787   & 1171  &  18.9 & 484  & 466  & 46.82  	  & 3.96    	   & 256 	& 23 	  & 1.7    \hspace{0cm} & 2.5\hspace{0cm} & * 	& *	& 	& +   	\\
  7166   &  998  &  15.9 & 140  & 121  & 67.58  	  & 4.05    	   & 304 	& 24 	  & 2.1    \hspace{0cm} & 2.2\hspace{0cm} & **      & *	& *	&    	\\
  7256   & 1091  &  16.9 & 271  & 239  & 48.11  	  & 3.21    	   & 226 	& 19 	  & 1.3    \hspace{0cm} & 1.8\hspace{0cm} & *  	& ** 	& n.a.	& +   	\\
  7489   & 1027  &  16.4 &  86  &  68  & 4.07  	  	  & 0.26    	   & 55		& 4.4 	  & 3.4    \hspace{0cm} & 2.7\hspace{0cm} &   	& n.a.	& n.a. 	&    	\\
  7506   & 2532  &  37.0 & 171  & 116  & 3.66   	  & 1.17    	   & 37 	& 6.7 	  & 4.1    \hspace{0cm} & 3.7\hspace{0cm} & *       & n.a.	& n.a.	&    	\\
  7704   & 937   &  14.8 &  93  &  61  & 6.49   	  & 0.34    	   & 58 	& 4.3 	  & 3.7    \hspace{0cm} & 9.4\hspace{0cm} &   	& * 	& 	&    	\\
  7989   & 1208  &  18.2 & 417  & 390  & 125.78 	  & 9.78    	   & 380 	& 34	  & 2.4    \hspace{0cm} & 2.7\hspace{0cm} & ***     & **	& *	& +   	\\
  8271   & 1128  &  17.7 & 166  &  76  & 10.37  	  & 0.78    	   & 89 	& 7.7 	  & 2.1    \hspace{0cm} & 3.7\hspace{0cm} & *  	& ***	& n.a.	& +   	\\
  8699   & 2521  &  36.7 & 387  & 368  & 10.22  	  & 3.26    	   & 100 	& 17 	  & 2.9    \hspace{0cm} & 4.6\hspace{0cm} &   	& 	& 	&    	\\
  8805   & 2384  &  34.5 & 129  &  82  & 15.45  	  & 4.34    	   & 137 	& 23 	  & 1.8    \hspace{0cm} & 3.0\hspace{0cm} &    	&  	& 	&    	\\
  8863   & 1791  &  27.2 & 391  & 367  & 14.44  	  & 2.52    	   & 148 	& 19 	  & 1.5    \hspace{0cm} & 1.6\hspace{0cm} & *  	& 	& *	&    	\\
  9133   & 3861  &  54.3 & 451  & 416  & 43.25  	  & 30.23    	   & 218 	& 57 	  & 1.9    \hspace{0cm} & 4.0\hspace{0cm} & **      & **	&  	&    	\\
  9644   & 6665  &  91.5 & 131  & 119  & 3.63   	  & 7.16    	   & 53		& 24 	  & 3.2    \hspace{0cm} & 4.4\hspace{0cm} &   	& 	& 	&    	\\
  10448  & 11351 & 153.8 &  99  &  85  & 1.76   	  & 9.82     	   & 40 	& 29 	  & 2.6    \hspace{0cm} & 2.0\hspace{0cm} &    	& 	& 	&    	\\
  11269  & 2581  &  38.3 & 415  & 344  & 31.32  	  & 10.86    	   & 161 	& 30 	  & 3.0    \hspace{0cm} & 6.1\hspace{0cm} & *  	& ***	& **	& +   	\\
  11670  & 775   &  12.7 & 342  & 325  & 28.05  	  & 1.08    	   & 158 	& 9.9 	  & 2.5    \hspace{0cm} & 2.5\hspace{0cm} & **      & 	& *	&    	\\
  11852  & 5846  &  80.0 & 329  & 305  & 19.55  	  & 29.6     	   & 137 	& 53 	  & 2.5    \hspace{0cm} & 4.2\hspace{0cm} & ** 	& 	& **	&    	\\ 
  11914  & 949   &  14.9 & 322  & 309  & 13.16  	  & 0.69   	   & 113 	& 8.1 	  & 3.0    \hspace{0cm} & 2.8\hspace{0cm} &   	& 	& 	&    	\\
  11951  & 1086  &  17.4 & 222  & 201  & 25.93  	  & 1.85    	   & 127 	& 10.7 	  & 4.4    \hspace{0cm} & 9.6\hspace{0cm} &   	& 	& 	& +   	\\
  12043  & 1007  &  15.4 & 190  & 176  & 20.96  	  & 1.18    	   & 143	& 10.8 	  & 2.4    \hspace{0cm} & 6.1\hspace{0cm} &   	& 	& 	&    	\\
  12276  & 5662  &  77.7 & 135  & 108  & 4.09   	  & 5.83    	   & 67 	& 26 	  & 2.2    \hspace{0cm} & 2.9\hspace{0cm} & *       & * 	& 	& +   	\\
  12713  &  295  &   5.7 & 147  & 101  & 10.79  	  & 0.08  	   & 84 	& 2.3 	  & 2.0    \hspace{0cm} & 4.3\hspace{0cm} & *  	& *	& *	&    	\\
  12815  & 4307  &  58.5 & 656  & 457  & 11.74$^\$$ 	  & 9.36$^\$$      & 103 	& 30 	  & 1.9    \hspace{0cm} & 2.6\hspace{0cm} & n.a.    & n.a.	& n.a.	& +   	\\\hline 
\end{tabular}
\smallskip
$^\#$ Average surface density $<$1 ~ $\msunpcdeux$
  everywhere. 
  \\ $^\dagger$ Total flux and \HI\ mass measured
  from $30\arcsec$ data.\\
$^\$$ Includes emission from UGC~12813.
\end{table}

\end{document}
